

#### Intergenerational mobility of immigrants in the Netherlands

Elk, R. van; Jongen, E.L.W.; Koot, P.; Zulkarnain, A.

#### Citation

Elk, R. van, Jongen, E. L. W., Koot, P., & Zulkarnain, A. (2024). *Intergenerational mobility of immigrants in the Netherlands. IZA Discussion Paper Series*. Bonn, Germany: IZA Institute of Labor Economics. Retrieved from https://hdl.handle.net/1887/3762000

Version:Publisher's VersionLicense:Leiden University Non-exclusive licenseDownloaded from:https://hdl.handle.net/1887/3762000

Note: To cite this publication please use the final published version (if applicable).



Initiated by Deutsche Post Foundation

# DISCUSSION PAPER SERIES

IZA DP No. 17035

Intergenerational Mobility of Immigrants in the Netherlands

Roel van Elk Egbert Jongen Patrick Koot Alice Zulkarnain

MAY 2024



Initiated by Deutsche Post Foundation

### DISCUSSION PAPER SERIES

IZA DP No. 17035

# Intergenerational Mobility of Immigrants in the Netherlands

Roel van Elk

CPB Netherlands Bureau for Economic Policy Analysis

**Egbert Jongen** Leiden University and IZA **Patrick Koot** Ministry of Social Affairs and Employment

Alice Zulkarnain CPB Netherlands Bureau for Economic Policy Analysis and IZA

MAY 2024

Any opinions expressed in this paper are those of the author(s) and not those of IZA. Research published in this series may include views on policy, but IZA takes no institutional policy positions. The IZA research network is committed to the IZA Guiding Principles of Research Integrity.

The IZA Institute of Labor Economics is an independent economic research institute that conducts research in labor economics and offers evidence-based policy advice on labor market issues. Supported by the Deutsche Post Foundation, IZA runs the world's largest network of economists, whose research aims to provide answers to the global labor market challenges of our time. Our key objective is to build bridges between academic research, policymakers and society.

IZA Discussion Papers often represent preliminary work and are circulated to encourage discussion. Citation of such a paper should account for its provisional character. A revised version may be available directly from the author.

ISSN: 2365-9793

IZA – Institute of Labor Economics

Schaumburg-Lippe-Straße 5–9	Phone: +49-228-3894-0	
53113 Bonn, Germany	Email: publications@iza.org	www.iza.org

### ABSTRACT

# Intergenerational Mobility of Immigrants in the Netherlands<sup>\*</sup>

A key measure of equality of opportunity is intergenerational mobility. Of particular interest is the extent to which children of immigrants catch up with natives. Using administrative data for the Netherlands, we find large gaps in the absolute income mobility of immigrants relative to natives (–23%), suggestive of large, persistent income gaps for future generations as well. Important drivers are differences in household composition and in personal incomes. However, we also uncover substantial heterogeneity by country of origin. Children of immigrants from China actually have higher incomes than natives, which is closely related to their educational outcomes.

JEL Classification:	D31, J15, J61, J62
Keywords:	intergenerational mobility, immigrants, Netherlands

**Corresponding author:** Alice Zulkarnain CPB Netherlands Bureau for Economic Policy Analysis Bezuidenhoutseweg 30 2594 AV The Hague The Netherlands E-mail: a.zulkarnain@cpb.nl

<sup>\*</sup> We are grateful to Spencer Bastani, Anne Gielen, Nathan Hendren, Maggie Jones, Jan Stuhler, Daniel Waldenstr'om, Dinand Webbink and participants of the CPB Workshop 'Inequality and Intergenerational Mobility' 2019 in the Hague, the Netherlands' Economists Day 2019 in Amsterdam, the KVS New Paper Sessions 2023 in the Hague, WAP 2023 in Bordeaux and ESPE 2023 in Belgrade for helpful comments and suggestions. This paper uses non-public microdata provided by Statistics Netherlands, made available for scientific research via remote access. The views and opinions expressed in this paper are those of the authors and do not necessarily reflect the official position of CPB Netherlands Bureau for Economics Policy Analysis, Leiden University or the Ministry of Social Affairs and Employment. The work on this paper was performed when Patrick Koot was employed at CPB Netherlands Netherlands Bureau for Economic Policy Analysis.

#### 1 Introduction

With income inequality rising in many high-income countries [see e.g. OECD, 2019, and the country studies for the Deaton Review (2023)], there is increasing concern for inequality of opportunity. Inequality of opportunity is important not only from the perspective of equity, but also matters for economic efficiency, as it may be an indication of a misallocation and underutilization of the human capital potential in society. An important indicator of equality of opportunity is intergenerational income mobility. Specifically, low intergenerational income mobility, when the income or characteristics of parents play a key role in the income of their children, is considered indicative of relatively poor equality of opportunity [Chetty et al., 2014, Blanden, 2019]. Conversely, when the income and characteristics of the parents play no role in the income positions of their children, this is indicative of high equality of opportunity, and income differences that exist for parents will disappear over future generations [Becker and Tomes, 1979].

Of particular interest is the intergenerational mobility of immigrants [Borjas, 1993, Aydemir et al., 2009, Hammarstedt and Palme, 2012, OECD, 2017, Abramitzky et al., 2021, Bratu and Bolotnyy, 2023] and ethnic minorities [Chetty et al., 2020]. Indeed, the extent to which the children of immigrants and ethnic minorities catch up with the children of natives or the majority, is an important indicator of how successful the educational and labor market institutions of a country are in generating equal opportunities for all groups in society. So far, the relatively small, but growing body of empirical research on intergenerational mobility of immigrants and ethnic minorities has focused on what can be considered cases of polar opposites when it comes to income inequality: Sweden [Hammarstedt and Palme, 2012, Bratu and Bolotnyy, 2023] and the US [Chetty et al., 2020, Abramitzky et al., 2021].<sup>1</sup>

Generally speaking (a more detailed comparison is given below), conditional on the income of parents, the incomes of children of immigrants and natives are quite comparable in Sweden [Bratu and Bolotnyy, 2023]. For the US, conditional on the income position of fathers, the income position of sons is actually better for immigrants than for natives [Abramitzky et al., 2021]. However, also for the US, conditional on the income of the parents, the income position of sons of Black Americans and Native-Americans is much lower than of sons of (non-Hispanic) white Americans, though there is almost no gap for daughters [Chetty et al., 2020]. Given these heterogeneous results, it is important to expand our understanding of

<sup>&</sup>lt;sup>1</sup>Meanwhile, the analysis of intergenerational mobility for the full population or by region, has proliferated, see e.g. Chetty et al. [2014], Heidrich [2017], Corak [2020], Deutscher and Mazumder [2020], Acciari et al. [2022], Kenedi and Sirugue [2023], and Manduca et al. [2024].

intergenerational mobility of immigrants and ethnic groups to other countries and regions.

In this paper, we study the intergenerational mobility of natives and immigrants in the Netherlands. We pay particular attention to the heterogeneity in results, by country of origin and by gender. Furthermore, we study the underlying factors driving the differences in intergenerational mobility across groups, both at the household level and at the individual level.

Our empirical methodology closely follows Chetty et al. [2020] and looks at the rank-rank relationship between the incomes of parents and that of their children. We rank children based on their average household income in 2017–2019 and rank their parents based on their average household income in 2003–2006. We study intergenerational mobility patterns by regressing the children's income rank on a constant and their parents' income rank, for natives and each of the selected immigrant groups. Such a rank-based approach works well from a statistical perspective, because the conditional expectation of the children's income is generally well approximated by a linear function of the parents' income [see Chetty et al., 2014, 2020, and the results below].

Using administrative data on income, education and labor market outcomes, we consider intergenerational mobility for children of native-born and immigrants, for immigrants from lower-income versus higher-income countries, and by country of origin for the largest groups of immigrants in the Netherlands from lower-income countries: Morocco, Türkiye, Surinam, the Antilles, Indonesia, and China. In our main analysis we use data on children born between 1983-1988. We measure children's income as their annual equivalized disposable household income averaged over 2017–2019, when the children are aged 29–36. We measure their parents' income as their annual equivalized disposable household income averaged over 2003–  $2006.^2$  when the children are aged 15–22. Equivalized disposable household income gives the economic resources available per person in the household, accounting for the number of adults and children in the household and for economies of scale in e.g. food and housing. We also consider the total wealth level of parents. Furthermore, we consider the share of children that are single, and, for those in couples, the share that is cohabiting with a partner with a background in a lower-income country, which may amplify the differences at the household level. For the children we also consider, by gender, personal primary income ('market income') and the factors that underlie primary income: the highest completed level of education, employment rates, hours worked and hourly wages.

 $<sup>^{2}2003</sup>$  is the earliest year for which we have integral income data in the Netherlands.

Our main findings are the following: First, we find large gaps in the absolute income mobility of immigrants relative to natives. At the lowest income rank of parents, the predicted gap for children is 9.3 percentiles (-23% relative to natives). And although we typically find lower relative income mobility for immigrants than natives (the coefficient on the parents income rank is 0.30 and 0.23, respectively), which implies that the gap relative to natives is smaller for children of parents with a higher income level, our findings suggest large and persistent gaps between immigrants and natives for future generations.

Second, focusing on immigrants from lower-income countries, we find that these differences are driven both by differences in household composition and differences in personal incomes. Children of immigrants, from the countries we study, tend to be single more often (and therefore more likely to be in a single-earner household), in particular when they have parents with a relatively low income level. Furthermore, for those in a couple, children of immigrants born in lower-income countries are more likely to have a partner with a background in a lower-income country as well, which amplifies gaps at the household level, as both earners likely experience negative income gaps. Regarding individual incomes, we find personal income gaps for both sons and daughters, which are closely related to gaps in the highest completed level of education, and is the sum of differences in employment rates, hours worked, and (typically to a lesser extent) hourly wages, all contributing to the gap.

Third, we also uncover substantial heterogeneity in the intergenerational mobility by country of origin. We find large and persistent negative gaps for children with parents from Morocco and Türkiye, and also from Surinam and the Antilles (despite their parents' proficiency in Dutch). For children with parents from Morocco and Türkiye this is driven by large gaps in personal incomes, and exacerbated because they are more likely to have a partner with a background in a lower-income country, who likely face a negative personal income gap themselves. For children with parents from Surinam and the Antilles, the negative gaps in personal incomes are somewhat smaller, but they are more likely to be single, in particular women. Our findings for children of parents born in Asian countries are more optimistic. We find almost no gap relative to natives for children with parents from Indonesia, and children with parents from China reach higher income levels than children of natives, in particular for parents with a relatively low income. The latter appears closely related to the high share of children in this group that complete a higher education (ISCED 6 or higher) relative to children of native-born.

Our paper contributes to the small but growing literature on intergenerational mobility of immigrants and ethnic minorities using linked income data for parents and children [Chetty et al., 2020, Abramitzky et al., 2021, Bratu and Bolotnyy, 2023]. The literature to date has focused on two polar opposite cases of income inequality and intergenerational income mobility: Sweden and the US. For the Netherlands, we find that at the lowest income rank of parents, the predicted gap for children in household income is 9.3 percentiles, which is much higher than the predicted gap of 3.2 percentiles for Sweden [Bratu and Bolotnyy, 2023]. Indeed, for many groups of immigrants from lower-income countries, the gaps in absolute income mobility relative ro natives are as large or larger than the gap in household income between children of Black Americans and children of (non-Hispanic) white Americans in the US [Chetty et al., 2020]. Accounting for the differences in relative income mobility as well (where the relative income mobility is typically somewhat lower for immigrants from lower-income countries), we predict large and persistent negative income gaps for descendants from most groups of immigrants from lowerincome countries, similar to what has been shown by Chetty et al. [2020] for Black and white Americans in the US. Furthermore, also consistent with the findings for children of Black Americans relative to children of white Americans in the US [Chetty et al., 2020], we find that differences in household income between children of immigrants from lower-income countries and natives are partially driven by a higher probability of being single and partially by differences in personal incomes. However, we also find that within couples, children of parents born in lower-income countries are more likely to have a partner with a background in a lower-income country as well, amplifying the differences at the household level. To the best of our knowledge, we are the first to uncover this channel.

Due to the richness of our dataset, we can also consider outcomes for both sons and daughters.<sup>3</sup> We find substantial gaps relative to natives for both sons and daughters of immigrants, which differs from the findings of Chetty et al. [2020] for children of Black Americans and children of white Americans in the US, where the difference in personal incomes appears to be driven solely by sons. Consistent with the findings of Chetty et al. [2020] and Abramitzky et al. [2021], we find smaller gaps for children of parents born in the Asian countries we study,<sup>4</sup> in particular from China, which appears closely related to their educational outcomes.

The paper is organized as follows. In Section 2, we first briefly discuss the most important immigration waves into the Netherlands that have determined the composition of immigrants that we study. Section 3 discusses the empirical approach

 $<sup>^{3}</sup>$ Abramitzky et al. [2021] focus on father-son pairs for the US, Bratu and Bolotnyy [2023] do not consider outcomes by gender for Sweden.

<sup>&</sup>lt;sup>4</sup>Hammarstedt and Palme [2012] finds a similar result for Sweden, using a slightly different methodology.

we use to estimate the intergenerational mobility patterns for natives and the immigrant groups. Section 4 describes the administrative income data, the estimation sample, and presents descriptive statistics for natives and different groups of immigrants. Section 5 presents the results for intergenerational mobility by country of origin, where we consider income at the household level, and household composition, individual incomes, education, and labor market outcomes by gender. Section 6 concludes the paper. Supplementary tables and figures are provided in the Supplementary Material.

#### 2 Immigrants in the Netherlands

We first provide some background information on the main immigration waves into the Netherlands since the late 1940s (see Nicolaas and Sprangers [2007] and Statistics Netherlands [2022] for a more detailed account).

The first immigration waves after the 1940s were the result of the decolonization process. Following the independence of Indonesia from the Netherlands in 1949, several large waves of Indonesian immigrants arrived in the late 1940s, the 1950s, and early 1960s. Immigration from Indonesia continued after the 19760s, but at a much slower pace than before.

The next immigration waves followed after a long period of labor market shortages in the 1950s and 1960s. In the 1960s, the Dutch government started actively recruiting workers from abroad, which resulted in immigration waves from Southern Europe, Morocco, and Türkiye. These waves were followed by family migration in the subsequent decades, mostly from Morocco and Türkiye, after the Dutch government allowed family reunification starting in 1974.

The next immigration waves were again related to the decolonization process. After Surinam became independent from the Netherlands in 1975, several migration waves to the Netherlands followed between 1975 and 1980. And as the economy of the Antilles went into a slump in the 1980s, there were several waves of migrants from the Antilles to the Netherlands. An important difference between the immigrants from the former Dutch colonies and other groups discussed above is that most of them were already proficient in Dutch upon arrival.

We focus our analyses on the groups of immigrants from Marokko, Türkiye, Surinam, the Antilles, and Indonesia, together with a more recent group of immigrants from China. However, let us briefly consider the subsequent waves of immigrants. Studying the intergenerational mobility of these more recent groups of immigrants is left for future research, when we can observe both the labor market outcomes for the parents and the children in these groups. In the 1990s, there was a substantial inflow of refugee immigrants, most notably from Somalia and former Yugoslavia. Subsequently, the 2000s saw a large inflow of labor immigrants from Eastern Europe, after the expansion of the European Union in 2004, in particular from Bulgaria, Poland, and Romania. In the 2010s there were several new waves of refugee immigrants, in particular from Syria, and since 2022 from Ukraine.

Official statistics on the share of individuals born outside of the Netherlands are available from the mid 1990s. In 1996, 8.3% of the population in the Netherlands was born outside of the Netherlands, and 7.8% of the population was the child of an immigrant. By 2022, 13.7% of the population in the Netherlands was born outside of the Netherlands, and 11.5% of the population was the child of an immigrant. The majority of the growth in the share of immigrants in the Netherlands is seen among the group that hails from lower-income countries, their share in the population has grown from 4.9% in 1996 to 8.0% in 2022. Furthermore, 2.6% of the population was the child of an immigrant from a lower-income country in 1996. By 2022 this share had gone up to 6.3%. The share of immigrants and their children is expected to increase further in the population in the coming decades [Stoeldraaijer et al., 2022]. Hence, studying the intergenerational mobility of immigrants, in particular among immigrants from lower-income countries, is also becoming increasingly important over time.

#### 3 Methodology

Following Chetty et al. [2014, 2020], we measure the level of intergenerational income mobility by assessing the correlation between the children's income rank and the parents' rank. This rank-based approach to characterize intergenerational mobility patterns starts by converting income to percentile ranks by ranking all children relative to other children in our estimation sample. We do the same for the parents. We then regress the income rank of the child on the income rank of the parents:

$$y_{i,t} = \alpha_r + \beta_r y_{i,t-1} + \epsilon_{i,t},\tag{1}$$

where  $y_{i,t}$  denotes child *i*'s percentile rank in the income distribution of children (generation *t*), and  $y_{i,t-1}$  is child *i*'s parents' percentile rank in the income distribution of the parents (generation t - 1). Furthermore, we estimate this relationship for each immigrant group separately, where *r* denotes the immigrant group. The income ranks are always based on the position in the entire estimation sample, also when we consider the outcomes for subgroups (by country of origin for example). The main advantage of this rank-based approach is that the relationship between a child's income rank and their parents' income rank is well approximated by a linear function, in contrast to when using a log-log specification.<sup>5</sup>

The estimated parameters  $\alpha_r$  and  $\beta_r$  can be interpreted as the absolute and relative rank mobility, respectively. The intercept  $\alpha_r$  indicates the average income rank of children in immigrant group r whose parents have the lowest income position. A high intercept implies that children of parents in the lowest income percentile still reach a relatively high income position. The slope  $\beta_r$  measures the correlation between a child's position in the income distribution and the income position of their parents. This parameter indicates how much the average child income rank increases if the income rank of the parents increases by one. It takes values between 0 (the highest relative income mobility) and 1 (the lowest relative income mobility). If the slope equals 0, all children (on average) end up at the same position in the income distribution, independent of the income position of their parents. If the slope equals 1 (and the constant is 0), each child (on average) ends up at the same position in the income distribution as their parents. Lower values of  $\beta_r$  thus imply higher relative intergenerational mobility, which means that a child's income is less related to the income of the parents.

In this paper, we focus on differences in the estimated intergenerational parameters (intercept and slope) between natives and different immigrant groups. If the intercept varies strongly across groups, this implies large income differences for children with parents in the lowest income percentile. If the slopes are more or less equal across groups, such income differences are more or less equal across the entire parental income distribution. If the slopes differ across groups, income differences between groups vary across the parental income distribution.<sup>6</sup>

We can also make a long-run prediction of the average income positions of future generations of natives and different immigrant groups, under the assumption that the intergenerational mobility parameters are stable across generations. Specifically, we can derive the steady-state average income ranks  $\bar{y}_r^{SS}$  for each group using

<sup>&</sup>lt;sup>5</sup>Earlier studies have estimated the intergenerational income elasticity by regressing log child income on log parent income [see e.g. Solon, 1999]. A disadvantage of this approach is that the relationship between log child income and log parent income is generally not linear, which leads to less robust estimates. Furthermore, estimates of the intergenerational income elasticity are also likely to be sensitive to the treatment of children with zero or small incomes [Chetty et al., 2014]. See Chetty et al. [2014] for a more detailed analysis of the relation between the two approaches.

<sup>&</sup>lt;sup>6</sup>Note that high relative mobility (a low value for the slope) within a migrant group could be the result of relatively good outcomes for children of low income families and/or relatively bad outcomes for children of high income families. Hence, we need to look at differences in both absolute and relative income mobility.

Equation (1) [see Chetty et al., 2020]:

$$\bar{y}_{r,t} = \bar{y}_{r,t-1} = \bar{y}_r^{SS} = \frac{\alpha_r}{1 - \beta_r}.$$
 (2)

We are not able to test the stability of the parameters across three generations for the Netherlands due to data constraints.<sup>7,8</sup> Hence, the predicted steady-state outcomes should be interpreted with the appropriate care.

#### 4 Data

We first discuss the selections we make to arrive at our estimation sample. Subsequently, we consider the definition of the variables in our analysis. We conclude with summary statistics of the estimation sample.

**Data Sets and Sample Selection** We use administrative data from Statistics Netherlands on the universe of individuals living in the Netherlands (in brackets are the abbreviatiated Dutch names of the data sets). Data on country of birth, age, gender, and position in the household for all members of the household, are taken from the municipal population registers (GBA). To these data, we link integral data on household income of the parents for the years 2003–2005 (IHI), personal and household income of the children for the years 2017–2019 (INPATAB and INHATAB), the highest completed level of education (Hoogsteopltab), and labor market outcomes (INPATAB and Spolisbus) of the children for the same years. We pool several years of data to limit the role of the business cycle and other transitory shocks. For a supplementary analysis, we also link data on parental wealth (VEHTAB). Because wealth data for all households in the Netherlands are only available since 2006, we pool these data for the years 2006–2008 (for the parents).

We make the following selections, we keep child-parents pairs with children who: i) live at home with their parents in 2003, ii) have parents that are younger than 65 in

<sup>&</sup>lt;sup>7</sup>The integral income data are only available from 2003 onwards. We have explored the Income Panel Study (*Inkomenspanelonderzoek*) of Statistics Netherlands to study intergenerational mobility for subsequent generations, which has personal identifiers since 1985. However, this is a sample of about 1.5% of the population, resulting in numbers of observations for immigrant groups that were too small.

<sup>&</sup>lt;sup>8</sup>Also in other countries, information on the development of incomes over multiple generations is limited. However, at least in two countries, Belgium and Sweden, there is administrative data to study the relationship between labor market outcomes of children and labor market outcomes of their grandparents. These analyses suggest that the gaps between groups persist over three generations [OECD, 2017]. Furthermore, Adermon et al. [2021] show that there is also a direct link from grandparents to grandchildren (and also from other extended family members), next to the link between parents and children.

2003, iii) are born between 1983 and 1988, iv) live independently from their parents in 2017–2019, and v) have strictly positive parental household income (including income from social benefits) in 2003–2005. Furthermore, a child is allocated to different groups based on the parents' country of birth. The country group is defined by the mother's country of birth. When the mother's country of birth is the Netherlands and the father's country of birth is not, the country group is defined by the father's country of birth. The groups of immigrant children consists of first and second generation immigrants. Statistics Netherlands considers third generation immigrants to be Dutch, we therefore include them and their second generation parents in the group of natives. We refer to this group of child-parents pairs as Dutch, natives, or as native-born parents and their children interchangeably.

We define parents as the head of the household, the child belongs to in 2003, and their partner. Hence, the child lives with these adults at the moment we define groups, but these may not be the child's biological parents.<sup>9,10,11</sup>

Because we select children born between 1983 and 1988, these children are between ages 15 and 22 in 2003–2005, and between ages 29 and 36 in 2017–2019. We choose the birth cohorts 1983—1988 so that children are old enough in 2017–2019, for their income to be a valid proxy of their lifetime income, but young enough to still be living with their parents in 2003. Measuring incomes at earlier ages could lead to biases in the estimated intergenerational income mobility parameters.<sup>12,13</sup>

Our estimation sample consists of 636,690 child-parents pairs, among which 536,348 pairs with native-born parents and 110,342 pairs with immigrant parents. The group of immigrant children consists of 63,588 individuals with a parent from a lower-income country and 36,754 with a parent from a higher-income country. Moving to individual countries of origin, we have 14,837 children of Surinam parents,

<sup>&</sup>lt;sup>9</sup>Hence the parental income is informative for the amount of available resources to the child when they are growing up. Since children do not necessarily grow up with the same parents during childhood (e.g. due to divorce), it is not a priori clear what the best measure of the parental income should be, to be the most informative for the available resources to a child during childhood.

<sup>&</sup>lt;sup>10</sup>We select children who live with their parents in 2003, which implies that parents living in an institutional household (such as homes for the elderly, nursing homes or other institutions) or a student household are excluded from the estimation sample.

<sup>&</sup>lt;sup>11</sup>For immigrants, linking children to their biological parents is not feasible due to missing information in the administrative files on biological linkages.

<sup>&</sup>lt;sup>12</sup>This bias is referred to as 'life-cycle bias' [Solon, 1999]. Chetty et al. [2014] investigate the life-cycle bias by estimating rank-rank slopes for different ages at which the child income is measured. They find that the estimates stabilize around age 30.

<sup>&</sup>lt;sup>13</sup>A potential drawback of this choice is that some of the (eldest) children in these birth cohorts have already left the household in 2003, and hence, are not included in the estimation sample. This might affect our estimates if the fraction of home-leavers differs across migrant groups. Estimates could be affected, for example, if children with lower income mobility stay in their parental household longer.

13,058 children of Turkish parents, 11,913 children of Moroccan parents, 7,132 children of Indonesian parents, 5,711 children of Antillean parents, and 1,584 children of Chinese parents.

**Variable Definitions** For the analysis of household income, we focus on equivalized disposable household income, both for parents and their children.<sup>14</sup> Equivalized disposable household income adjusts disposable household income for differences in household composition, using equivalence scales, so that the equivalized disposable household income is the single-person household equivalent of the disposable income of the household. In this way, households can be compared in their level of resources available per person. As a robustness check, we also consider non-equivalized disposable household income [as in e.g. Chetty et al., 2020]. The wealth variable we use in a supplementary analysis is net total household wealth, including net housing wealth.

For the children, we also consider a number of individual outcomes, which we study by gender. We show results for personal primary income, which is income from wages and profits ('market income' before taxes and transfers), employment (a binary indicator that is one when the child has non-zero wages or profits), hours worked (when employed), and hourly wages. The latter two outcomes are only available for employees. Furthermore, we consider whether the child has completed an intermediate or higher education, defined as having completed ISCED 3 or higher, and whether the child has completed a higher education, defined as having completed ISCED 6 or higher.<sup>15</sup>

Summary Statistics Table 1 presents selected summary statistics for the estimation sample.<sup>16</sup> Starting with the parents in Panel A, we see that the median equivalized disposable household income of immigrant parents (in 2003–2005) is much lower than that of native-born parents.<sup>17</sup> This is particularly true for the (total) group of immigrants from lower-income countries,<sup>18</sup> and for the individual

<sup>&</sup>lt;sup>14</sup>Disposable household income consists of annual income from wages, profits, wealth, social insurance benefits and welfare benefits, minus income taxes and social security contributions, plus transfers.

<sup>&</sup>lt;sup>15</sup>We do not use data on the highest completed level of education of the parents, because these data have many missings (whereas we have almost complete coverage for the children).

<sup>&</sup>lt;sup>16</sup>A larger set of summary statistics are given in Table A.1 and A.2 in the Supplementary Material.

<sup>&</sup>lt;sup>17</sup>Figure A.1 in the Supplementary Material shows where the parents of the different countries of origin are located in the overall income distribution of parents.

<sup>&</sup>lt;sup>18</sup>Our definition for lower- and higher-income countries is based on the earlier classification used by Statistics Netherlands for so-called Western and Non-Western countries. Under this classification Indonesia is classified as Western [Statistics Netherlands, 2018a].

Panel A: Income and household composition parents	(	Higher-income country	Lower-income country	Moroccan	Turkish	Surinamese	Antillean	Indonesian	Chinese
Median equivalized household income 25,073	3 19,466	23,361	17,610	15,665	16,902	19,827	19,347	25,124	17,759
Median stand. household income rank 54	4 26	46	18	12	16	28	26	54	19
Share single parent 0.14	4 0.31	0.25	0.34	0.20	0.24	0.50	0.49	0.24	0.20
Panel B: Income and household composition children									
Median equivalized household income 31,014	4 26,867	29,692	25,112	22,148	25,051	25,167	23,987	30,598	33,122
Median stand. household income rank 53	3 36	47	30	20	30	30	26	51	61
Median personal income 34,637	7 29,901	33,013	27,955	23,829	27,082	28, 320	26,767	34,137	38,831
Median personal income rank 52	2 41	48	36	27	34	37	34	51	62
Share single or single parent 0.30	0 0.40	0.37	0.42	0.42	0.26	0.50	0.50	0.37	0.42
Panel C: Education and labor market outcomes children									
Share higher educated 0.44	4 0.38	0.45	0.34	0.28	0.27	0.33	0.31	0.50	0.64
Share employed 0.94	4 0.86	0.91	0.83	0.72	0.82	0.86	0.85	0.93	0.95
Mean hours worked per week 30.8	8 26.4	29.3	24.7	20.1	23.8	25.7	25.0	30.4	32.5
Mean hourly wage 18.8	8 18.1	18.6	17.9	17.8	17.3	17.6	17.6	18.7	19.7
Median hourly wage rank 52	2 44	48	42	43	39	41	41	48	56
Observations 536,348	8 100,342	36,627	63,588	11,960	13,058	14,864	5,720	7,140	1,589

is determined by the set of individuals living in the same household. Higher educated is defined as ISCED 6 or higher. Employed is defined as having nonzero wages and profits. Hours worked and hourly wages are for employees only (so excluding the non-employed and the self-employed). The hourly wage ranks are the ranks in the full estimation sample of children. Outcomes for parents are averages for 2003-2005, outcomes for children are averages for 2017-2019. Observations are the number of child-parental household pairs. countries of origin reported in this table (parents from Indonesia being the notable exception). This is also reflected in the household income ranks of the parents. Part of the differences are due to the larger share of single parents among immigrants, for whom the share of single parents is often more than twice as high for immigrants from lower-income countries as for native-born parents, and the highest for parents from the Antilles and from Surinam.

Turning to the outcomes for children (in 2017–2019) in Panel B, the relative (percentage) differences in the median equivalized disposable household income between children of native-born and children of immigrants are smaller than for their parents. However, sizable differences remain for most groups. The notable exception are children of Chinese parents, who have a higher median household income than children of native-born, whereas their parents on average still had substantially lower median household income than the native-born. The differences in the median household income soft the children are also reflected in the corresponding median household income ranks. The differences in household incomes also reflect the differences in personal primary incomes for children. The differences in the share that is single or a single parent is less pronounced for the children than for the parents. However, among those with parents from Surinam and the antilles, the relatively high share of singles may be an important factor behind the lower median rank for equivalized disposable household income than for personal primary income.

Panel C gives the educational and labor market outcomes for the children. Among children of immigrants, the share that has completed a higher education tends to be lower than among the children of native-born (though the share is similar to that of children of native-born, among children of immigrants from higherincome countries). The share that has a completed higher education is roughly 16 percentage points lower for children of parents from Türkiye and Morocco, and 11-13 percentage points lower for children of parents from Surinam and the Antilles, compared to children of native-born. On the flip-side, children of parents from Indonesia and China are respectively 6 and 20 percentage points more likely to have completed a higher education than children of native born.

Compared to children of native-born, children of immigrants tend to have lower employment rates and work fewer hours per week, although children of Indonesian and Chinese parents have similar employment rates and hours worked. The differences in hourly wages between children of immigrants and native-born are less pronounced in relative terms than differences in employment rates and hours worked, and hence seem to play a less important role in the differences in personal primary incomes.

It is noteworthy that, while children of immigrants from higher-income countries are equally likely as their counterparts with native-born parents to be higher educated, there are (small) negative differences in their employment rates, hours worked and hourly wages, leading to lower personal and household incomes. A similar pattern is visible for children of parents from Indonesia and China relative to children of native-born, who despite greater shares of higher education have similar or modestly higher incomes. Thus, higher education does not seem to automatically translate into higher income for immigrant children, suggesting that there is inequality in the returns to education.

These descriptive statistics are already informative, however, they are not conditional on parental income. More relevant for the long-run/steady-state differences are the differences conditional on parental income, which also allow us to study differences in outcomes for children growing up in immigrant households with relatively low or high financial resources. We turn to these results next.

#### 5 Results

We first consider the intergenerational mobility in household income for the pooled sample of all child-parents pairs, for natives and immigrants, and for immigrants from lower- and higher-income countries. Next, we look more closely into the intergenerational mobility by country of origin for the largest groups of immigrants from lower-income countries. We consider the heterogeneity in intergenerational mobility in household income by country of origin (for which we also present a number of robustness checks) and the corresponding predicted steady-state differences in household incomes. Also, we consider differences in household composition for children (the share that is single or a single parent, and the share that has a partner with a background in a lower-income country) and individual incomes (primary/market income), by parental income, where we focus on outcomes by gender for each country of origin. Finally, we consider potential factors that drive the differences in individual incomes, i.e. the highest level of completed education, employment shares, hours worked, and hourly wages ranks of employees.

#### 5.1 Household Incomes

**Native-Born and Immigrants** Figure 1(a) plots the mean (equivalized disposable) household income rank of the children (vertical axis) against the household

income rank of the parents (horizontal axis), both for the pooled sample and for natives and immigrants separately. All figures also show the best linear fit, and we include the estimated constant (absolute mobility coefficient) and slope (relative mobility coefficient) for each group in the legend.

In general, there tends to be an almost linear relationship between the average child income rank and the income rank of the parents. Indeed, the linear relationship seems to hold, except at the very bottom and top of the parental income distribution. A plausible explanation for these relatively favorable outcomes for children at the very bottom and top of the distribution, is the relatively high parental wealth at these ends of the distribution, see Figure A.2 in the Supplementary Material, which plots parental wealth rank against the parental income rank.<sup>19</sup> However, the intergenerational mobility results are very similar when we exclude the bottom 5% and top 5% of the parental income distribution, as we will see below in the robustness checks. Furthermore, note that the immigrant group is a bit more noisy around the linear approximation, which is likely to be due to the smaller sample size than for natives.<sup>20</sup>

In the pooled sample we find a relative income mobility of 0.26. This implies that a 10 percentile increase in the parents' income rank is associated with an average increase in the child's income rank of 2.6 percentiles. Chetty et al. [2020] report an estimated relative income mobility of 0.35 for the United States. This indicates that the incomes of children in the Netherlands are less dependent on the incomes of their parents than in the United States. Furthermore, when we use non-equivalized household income for parents and children instead, as in Chetty et al. [2014, 2020] (and most other studies), the relative income mobility for the pooled sample drops further down to 0.22<sup>21</sup> This estimate of the overall relative income mobility (for the joint group of natives and immigrants) in the Netherlands is also lower than the 0.30 that Kenedi and Sirugue [2023] estimate for France, somewhat below the 0.24and 0.25 that Corak [2020] and Acciari et al. [2022] estimate for Canada and Italy, respectively, and similar to the 0.22 that Deutscher and Mazumder [2020] estimate for Australia. However, the estimate for the Netherlands is still somewhat higher than in selected Scandinavian countries, like the 0.18 that Chetty et al. [2014] infer for Denmark and the 0.20 that Heidrich [2017] estimates for Sweden.

 $<sup>^{19}\</sup>mathrm{Figure}$  A.3 plots the average wealth level (instead of the rank) of parents against the income rank of the parents.

<sup>&</sup>lt;sup>20</sup>Statistics Netherlands prohibits presentation of means for cells with fewer than five households (individuals), due to identification risks. Those means are omitted from the scatter plots, but all regression analyses include all households (individuals).

<sup>&</sup>lt;sup>21</sup>See Figure A.4 in the Supplementary Material for the outcomes using non-equivalized disposable household income for parents and children by country of origin.



Figure 1: Relationship Parent–Child Household Income Ranks: Natives and Immigrants

*Notes:* Estimated linear relationship between the equivalized disposable household income rank of the parents and that of the children using Equation (1).

Children of native-born parents have a higher absolute mobility in the Netherlands ( $\alpha_r$  of 39.8) than children of immigrants ( $\alpha_r$  of 30.5). They also have a higher relative mobility ( $\beta_r$  of 0.23) than children of immigrants ( $\beta_r$  of 0.30), hence the income of immigrant children depends more their parents' income than is the case for children of native-born in the Netherlands. Across the entire parental income distribution, children of immigrants have a lower mean income rank than the children of native-born, with larger gaps at lower parental incomes than at higher incomes. There are only a handful of studies considering intergenerational mobility separately for natives and immigrants, for Sweden and the US. Bratu and Bolotnyy [2023] find qualitatively similar results for natives and immigrants in Sweden as we find for the Netherlands, with higher absolute mobility for natives than for immigrants ( $\alpha_r$  of 41.3 and 38.2, respectively) and also higher relative mobility for natives than for immigrants ( $\beta_r$  of 0.18 and 0.20, respectively). However, the differences in absolute and relative income mobility they find between natives an immigrants are much smaller than in the Netherlands. The picture for the US is quite different, where Chetty et al. [2020, Figure III in the Online Appendix] show that children of mothers born outside of the US have higher absolute and relative income mobility (higher  $\alpha_r$ , lower  $\beta_r$ ) than children of mothers born in the US. Abramitzky et al. [2021] show similar results for sons of fathers born outside the US relative to sons of fathers born

in the US. Explaining these heterogeneous differences in intergenerational mobility between natives and immigrants across countries is an interesting topic for future research, where differences in composition, policies, and societal factors likely play a role. The analysis below may also be informative about the factors that underlie these cross-country differences between natives and immigrants.

As a first step, Figure 1(b) splits the group of immigrants into immigrants from higher-income and lower-income countries.<sup>22</sup> We find that the gaps are much larger for children of parents from lower-income countries than from higher-income countries, suggesting heterogeneity in the intergenerational mobility of immigrants relative to natives across countries of origin. In the remainder of the paper we will focus on the largest groups of immigrants from lower-income countries.

Selected Countries of Origin Figure 2 gives the intergenerational mobility for immigrants from Morocco and Türkiye (panel (a)), Surinam and the Antilles (panel (b)), and Indonesia and China (panel (c)), and as the reference also for children of native-born. We find substantial heterogeneity in the intergenerational mobility among these groups.

Absolute income mobility is much lower for children of parents from Morroco and Türkiye ( $\alpha_r$  of 25.8 and 31.3, respectively) than for children of native-born parents ( $\alpha_r$  of 39.8). To put this in perspective, the gap in absolute income mobility for children of parents from Morocco relative to children of native-born parents is larger than the gap in absolute income mobility between children of Black and white Americans in the US [25.4 and 36.8, respectively, Chetty et al., 2020, p. 732]. Relative income mobility for children of Moroccan parents is quite similar to that for children of native-born parents, and slightly higher for children of Turkish parents.<sup>23</sup> Hence, there is a sizable gap in incomes for children with a parent from Morocco or Türkiye relative to children of native-born parents across the entire parental income distribution (with the exception of children with parents from Morocco at the very top of the parental income distribution).

Absolute income mobility is also much lower for children of parents from Surinam and the Antilles ( $\alpha_r$  of 26.9 and 20.4, respectively) compared to children of nativeborn parents ( $\alpha_r$  of 39.8), despite their parents' higher proficiency in the Dutch language compared to other immigrants from lower-income countries. Hence, there

<sup>&</sup>lt;sup>22</sup>We follow the earlier classification as defined by Statistics Netherlands Statistics Netherlands [2018a]. The group of higher-income countries consists of countries in Europe, North-America, Oceania, Indonesia and Japan. The group of lower-income countries consists of countries in Africa, Latin-America, Asia (excluding Indonesia and Japan) and Türkiye.

<sup>&</sup>lt;sup>23</sup>Chetty et al. [2020] find lower relative income mobility for children of Black and white Americans, 0.28 and 0.32 respectively [Chetty et al., 2020, p. 732].



Figure 2: Relationship Parent-Child Household Income Ranks: Selected Countries

*Notes:* Estimated linear relationship between the equivalized disposable household income rank of the parents and that of the children using Equation (1).

is a large negative gap at lower levels of parental incomes for these groups, relative to children of native-born parents. The relative income mobility for children of parents from Surinam and the Antilles is lower (the slope is higher) than that for children of native-born ( $\beta_r$  of 0.29 and 0.42, respectively). Hence, for children of parents with a relatively high income, the gap is much smaller, in particular for children with parents from the Antilles (they reach higher income levels than children of native-born parents at the very top of the income distribution of parents).

The results are quite different for children with parents from Indonesia and

China. For children with parents from Indonesia, absolute income mobility is comparable to that of children of native-born ( $\alpha_r$  of 37.9 vs. 39.8 for children of nativeborn), but is notably higher for children with parents from China ( $\alpha_r$  of 51.8). Even at the lowest parental income rank, the children of Chinese parents make it into the top half of the income distribution of all children. Also, whereas the relative income mobility of children with parents from Indonesia ( $\beta_r$  of 0.25) is quite comparable to that of the children of native-born, the relative income mobility of children with parents from China is again higher ( $\beta_r$  is lower at 0.15) than of children of nativeborn. Therefore, mean income ranks of children with parents from Indonesia are quite comparable to that of the children of native-born all across the income distribution, whereas the mean income ranks for children with parents from China are always above those of the children of native-born, and considerably higher than for children of native-born at the lower end of the parental income distribution. This latter finding is consistent with the findings of Chetty et al. [2020] for the children of Asian parents.

**Upward and Downward Mobility** The income gaps between the children of native-born and immigrants from lower-income countries can arise from differences in upward mobility and/or downward mobility. Table 2 gives the transitions from the bottom and top quintiles of the parental income distribution to the bottom and top quintiles of the income distribution of the children, by country of origin.

Among children with parents in the bottom quintile, 13% of children of nativeborn move up to the top quintile, while only 5% of children with a parent from the Antilles makes it into the top quintile, and the shares are also much lower for children of parents from Morocco, Türkiye or Surinam. Among children of parents from Indonesia, the transition probability is quite comparable to the children of native-born. However, of the children with a Chinese parent, 27% go from the bottom quintile of the parental income distribution to the top quintile of the child income distribution, suggesting significant upward mobility for this group.<sup>24</sup>

Among children with parents in the top quintile, only 10% of children of nativeborn fall down into the bottom quintile, while 29% of children of Turkish parents do. This probability is also higher for children with parents from Morocco, Surinam, or the Antilles than for children of native-born. Downward mobility is also quite similar to that for children of native-born, for children of Indonesian and Chinese

 $<sup>^{24}</sup>$ For the US, Chetty et al. [2020, Table 1] find that 11% of children of white parents in the bottom quintile make it to the top quintile, and this share is much lower for children of Black parents (6%), but much higher for children of Asian parents (27%).

Nat	ive-born	Moroccan	Turkish	Surinamese	Antillean	Indonesian	Chinese
(parents + their of a bar)	children)						
P(Child Q1   Parent Q1)	0.31	0.53	0.43	0.50	0.61	0.34	0.21
P(Child Q5   Parent Q1)	0.13	0.06	0.08	0.07	0.05	0.13	0.27
P(Child Q1   Parent Q5)	0.10	0.28	0.29	0.20	0.15	0.11	0.11
$P(Child~Q5 \mid Parent~Q5)$	0.31	0.25	0.17	0.23	0.34	0.35	0.46

Table 2: Transition Matrix: Upward and Downward Mobility by Immigrant Group

NOTES: Child income is the mean of 2017-2019 equivalized disposable household income (when the child is between 29 and 36 years old), while parental income is the mean of equivalized disposable household income in 2003-2005. Children are assigned percentile ranks relative to all other children in their birth cohort, while parents are ranked relative to all parents with children in the same birth cohort. Q1 and Q5 refer to the first and fifth quintiles of the relevant income distribution.

parents.<sup>25</sup> Hence, the differences in intergenerational mobility seem to be driven by both differences in upward and in downward mobility. The relative favorable outcomes for children with parents from China seem to be driven for the most part by relatively high upward mobility.

Steady-State Predictions The estimated parameters can also provide insight into the evolution of income disparities in the long run. Under the assumption that the absolute and relative income mobility parameters per group are stable over generations, we can make a prediction of the steady-state income gaps in the long-run. Plugging the estimated parameters presented in Figure 2 into Equation (2) gives us the steady-state mean income ranks for natives and for each of the immigrant groups. Figure 3 gives the mean household income rank of the parents, the mean household income rank of the children, and the predicted steady-state mean income rank, for each group.

The resulting steady-state income ranks suggest substantial income differences between natives and descendents of immigrants that persist in the long run. The steady-state income rank for natives is 52, while the steady-state income ranks for individuals with a Turkish, Moroccan, Surinamese, and Antillean background are 39, 34, 38, and 35, respectively. The persistence of these income gaps can be explained by the finding that these immigrant children have lower incomes than children of native-born even if their parents have similar income levels (see Figure 2). The steady-state prediction for future generations with a background in Indonesia is 51, and thus again very close to natives, and 61 for future generations with a background in China. Overall, differences between the average steady-state income rank and the (current) income rank of the children are small. Due to the high

 $<sup>^{25}</sup>$ For the US, Chetty et al. [2020, Table 1] find that 12% of children of white parents in the top quintile drop to the bottom quintile, and this share is somewhat higher for children of Black parents (14%), and quite comparable for children of Asian parents (11%).



Figure 3: Mean Parents, Mean Children and Steady-State Mean Income Ranks

relative income mobility of most groups, the income distributions for the different groups converge relatively quickly to their steady state. Hence, when the parameters of absolute and relative income mobility continue to hold for subsequent generations, existing gaps between the different groups will persist for future generations.

**Robustness Checks** Next, we consider how robust our findings are to different estimation samples or income measures. The results are shown in Table 3, where row 1) gives the baseline results.

In row 2) we restrict the sample to parents with an income rank between the 5th and 95th percentile (for parents outside this range, household income may work less well as a proxy for financial resources, as many of them have above average wealth levels, see Figure A.2 in the Supplementary Material). The results from this sample are very similar to the baseline results. In row 3) we restrict the age range of the parents to 35 - 55. This increases the homogeneity of the included households and reduces the risk of measuring the parental income too late in their career to be informative for lifetime income. Again, the results are very similar to the baseline results. In row 4) we investigate whether it matters if a child grows up in a single- or two-parent household. Especially children with parents from Surinam or the Antilles are more likely to grow up in single-parent families (see Table 1). We explore how these differences across migrant groups affect the results, by estimating the model on a sample restricted to two-parent households. This hardly affects the estimated parameters for the native-born, Moroccan, Turkish, Indonesian, and Chinese groups. However, children of parents from Surinam or the Antilles have

1	Native-born	Moroccan	Turkish	Surinam.	Antillean	Indones.	Chinese
(parents + the	eir children)						
Estimated intercept and slop	e						
1) Baseline	39.8; 0.23	25.8; 0.23	31.3; 0.19	26.9; 0.29	20.4; 0.42	37.9; 0.25	51.8; 0.15
2) Excl. bottom and top $5\%$	39.1; 0.24	26.7; 0.21	31.7; 0.19	26.7; 0.29	20.3; 0.41	37.1; 0.25	52.1; 0.16
3) Age of parents $35-55$	40.2; 0.23	25.6; 0.25	31.5; 0.20	27.5; 0.28	20.8; 0.42	38.0; 0.25	51.9; 0.15
4) Two-parent households	41.1; 0.22	26.4; 0.23	32.1; 0.18	31.0; 0.25	24.4; 0.37	39.4; 0.23	51.6; 0.16
5) Children born in the NL	39.8; 0.23	26.6; 0.25	32.0; 0.19	26.9; 0.30	24.5; 0.38	37.8; 0.25	53.9; 0.12
6) Pre-tax income	40.2; 0.22	29.4; 0.20	34.7; 0.20	28.6; 0.26	22.8; 0.38	38.3; 0.22	48.7; 0.16
7) Non-equiv. income	41.4; 0.20	29.9; 0.16	37.6; 0.17	29.9; 0.23	24.8; 0.33	39.3; 0.19	47.9; 0.14

Table 3: Robustness Checks

*Notes:* This table shows the intercept and slope from rank-rank regressions by group. 1) The baseline analyses regress the rank of the children's equivalized disposable household income on that of the parents. The sensitivity analyses are run on: 2) the subsample without the top and bottom 5% of the parental distribution, 3) the subsample of households with parents between ages 35 and 55, 4) the subsample where the parental households consisted of two parents, 5) the subsample of children that were born in the Netherlands, 6) the baseline sample but with rank-rank regressions of pre-tax income, 7) the baseline sample but with rank-rank regressions of non-equivalized income.

higher estimated intercepts and lower slopes when restricted to two-parent households, indicating that lower mobility for their children (in the baseline results) is partially driven by the larger share of single-parents (and thus single-earner households) in these groups. In row 5) of Table 1, we address differences between firstand second-generation children. First-generation children (children born outside of the Netherlands) have lived in the Netherlands for a shorter period (on average) compared to second-generation children (children born in the Netherlands to parents born abroad), which may influence the results. We explore this by including only second-generation children and their parents in the analysis. If we focus the analysis solely on second-generation children, the estimated absolute mobility is slightly higher for the groups with a Moroccan, Turkish, or Chinese background. The difference with the baseline results is the largest for children with parents from the Antilles, among whom the share of first-generation children is significantly larger compared to the other groups (see Table A.2).

Using alternative income measures has a limited effect on the results. In row 6) we use pre-tax household income instead of disposable household income, which yields comparable results. In row 7) we use non-equivalized (disposable) household income instead of equivalized household income (see Figure A.4). This tends to yield higher estimated intercepts and lower slopes for most of the immigrant groups.<sup>26</sup> It

 $<sup>^{26}</sup>$ Results from a log-log model of equivalized (real) household income typically show results qualitatively similar to those from our baseline model. The slopes for groups with a background in Morocco and Indonesia are 0.20, which is very similar to the 0.18 for native-born, but with lower intercepts (8.01 and 8.29, respectively, compared to 8.45 for the native-born). These slopes are to be interpreted as elasticities (for an elasticity of 0.18, children whose parental income is 10% higher, their income is 1.8% higher). The groups with a background in Surinam or the Antilles

should be noted that the differences in absolute mobility are noticeably larger for the groups from Turkiye and China, when using equivalized disposable household income (baseline) instead of non-equivalized disposable household income. This means that equivalizing leads to lower absolute mobility for the group from Turkiye and to higher absolute mobility for the group from China.

The set of sensitivity analyses shows that the main findings are robust to the use of different samples and different income concepts. Indeed, all robustness checks point to large and persistent negative income gaps between natives and immigrants from Morocco, Türkiye, Surinam, and the Antilles, small gaps between natives and immigrants from Indonesia, and positive gaps in income for children of Chinese parents relative to children of native-born.

#### 5.2 Household Composition and Personal Incomes

Chetty et al. [2020] show that a large part of the differences in household income between children of Black and white Americans is driven by a larger share of single parents (and non-parents) among children of Black parents.<sup>27</sup> Income differences are primarily driven by the fact that coupled households may be dual earner households, noting that we use equivalized disosable household income, which accounts for returns to scale in terms of e.g. food and housing, which can dampen income gaps. Chetty et al. [2020] also show that a smaller part is driven by differences in personal income, with small differences for women and larger differences for men. In this subsection we consider the differences in household composition, where next to being single or not, we also consider whether the child has a partner with a migration background in a lower-income country, and differences in personal incomes by gender.

**Share Single** Figure 4 shows that for most levels of parental income, children of immigrants from lower-income countries tend to be single more often than children of native-born. (This is similar to the findings of Chetty et al. [2020] who find substantial gaps in marriage rates between children of Black and white parents in the US across the income distribution of parents.) These groups will therefore have higher shares of single-earner households, leading to lower household income. The

have steeper slopes (0.26 and 0.36 respectively) than the native-born, with lower intercepts (7.48 and 6.53). The results for the group with a Chinese background again shows a flatter slope (0.11) with a higher intercept (9.28) than natives. Perhaps somewhat surprisingly, the results for the group with a background in Türkiye also show a flatter slope (0.13) than natives.

<sup>&</sup>lt;sup>27</sup>The difference in marriage rates is in the order of 32 to 34 percentage points [Chetty et al., 2020, Figure IV].

notable exception in our study are children with parents from Türkiye, who are typically less likely to be single than children of native-born.

The difference in the share of singles between immigrants and natives is particularly pronounced for children of parents with a relatively low income, with children of parents from Türkiye again being the notable exception. Indeed, it is striking to see how the share of singles among children of native-born is relatively flat across the income distribution of the parents, whereas the share of singles is typically declining with the level of parental income for children of immigrants. The difference is particularly strong for women with parents from Surinam or from the Antilles, and for men with parents from Morocco. Hence, part of the differences in equivalized disposable household income for children at the lower end of the parental income distribution stem from differences in the share that are single.

Share with Partner from a Lower-Income Country Figure 5 shows the share of children in couples that have a partner that is a first or second generation immigrant from a lower-income country. This may amplify differences at the household level, when e.g. both partners have a greater likelihood to have a lower income than children of native-born. We see that in particular children with parents from Morocco or Türkiye, but also children with parents from China, are more likely to have a partner with a migration background in lower-income countries than children of native-born. The difference is still pronounced, but somewhat smaller for children with parents from Surinam or the Antilles, and relatively small for children with parents from Indonesia. Interesting to note is the relationship with parental income. The higher the income of the parents, the lower the likelihood that the child's partner is a first or second-generation immigrant from a lower-income country. This is particularly true for children of parents from the Antilles. This also partially explains the steeper slope in household income rank for some of the groups of immigrants in Figure 2, in particular for the children with parents from the Antilles.

**Personal Primary Incomes** Besides differences in household composition, differences also arise from differences in personal incomes. Because the personal primary income (market income) ranks are very similar to the personal gross income (market income plus benefits) ranks, we only discuss one of these. We focus on ranks of primary incomes. The relationships between the ranks of personal gross income and parents' household income are provided in Figure A.6 in the Supplementary Material.



Figure 4: Relationship Household Income Parents – Share Children Single

*Notes:* Estimated linear relationship between the household income rank of the parents and the probability that the child is single.



Figure 5: Relationship Household Income Parent – Partner Lower-Income Country

*Notes:* Estimated linear relationship between the household income rank of the parents and the probability that a child in a couple has a partner with a migration background in a lower-income country.



Figure 6: Relationship Household Income Parents – Primary Incomes Children

*Notes:* Estimated linear relationship between the household income rank of the parents and the labor income rank of the children using Equation (1).

Figure 6 shows, by gender, the relationship between the equivalized disposable household income rank for parents and the (mean) personal primary income ranks for children. For men, we find substantially lower primary income ranks for children of parents from Morocco, Türkiye, Surinam, and the Antilles than for children of native-born. These differences become smaller for sons of Surinamese and Antillean parents with a higher income level (outcomes at the 25th and 75th percentile of the parental income distribution are given in Table 4 below), which also partly explains the steeper slope for equivalized disposable household income in Figure 2. Primary income gaps are much smaller for sons of Indonesian or Chinese parents, relative to sons of native-born, than they are for the other groups.

The results for women are more diverse. Women with a parent from Morocco or Türkiye, like men, also have substantially lower primary income ranks than women with native-born parents, and this difference actually increases somewhat when we consider parents with a higher income level. Women with a parent from Surinam or the Antilles have mean primary income ranks that are actually not that different from women with native parents, with a slightly steeper slope for women of parents from the Antilles. Women with parents from Indonesia also have quite similar mean primary income ranks as women with native-born parents. However, women with a parent from China have much higher primary incomes than women with native-born parents, especially daughters of parents with a relatively low income.

Hence, the differences in household income between children of native-born and immigrants are driven, in part, by both men and women among those with a background in Morocco and Türkiye (lower incomes), mostly by men for children of parents from the Antilles and Surinam (lower incomes), and mostly by women among those with a background in China (higher incomes). Differences between children of Indonesian parents and children of native-born are small for both men and women. These results differ from the findings of Chetty et al. [2020] for Black and white children in the US, who find that the results are driven primarily driven by differences for men, with only small differences for women.

#### 5.3 Education and Labor Market Outcomes

The differences in children's primary incomes are the result of differences in labor market outcomes, in which differences in education are likely to play an important role. Below we consider the differences for the highest completed level of education, employment rates, hours worked per week, and hourly wages, using the same framework as before. The (predicted) outcomes are summarized in Table 4, which gives the predicted outcomes at the 25th and 75th percentile of the equivalized disposable household income rank of the parents using Equation (1). The figures, including the estimated absolute and relative mobility parameters, are given in the Supplementary Materials.

**Highest Completed Level of Education** First we consider the share of children that has completed an intermediate or higher level of education (ISCED 3 or higher). Men with parents from Morocco, Türkiye, Surinam, and the Antilles are less likely to have completed this level of education than children of native-born. Figure A.7 (in the Supplementary Materials) shows that this negative gap shrinks at the top of the parental income distribution for sons of parents from Morocco, and even turns positive for sons of parents from the Antilles. Differences for men with parents from Indonesia and natives are again small, while men with Chinese parents are more likely to have completed at least an intermediate level of education than sons of native-born across the entire parental income distribution. For daughters of immigrants, gaps in having completed an intermediate or higher level of education are small relative to daughters of native-born.<sup>28</sup> However, women with parents from China with a relatively low income are much more likely to have completed an intermediate or higher level of an intermediate or higher level of an intermediate or higher level of native-born.

Next, we consider the share of children that has completed a higher level of education (ISCED 6 or higher).<sup>29</sup> There are some marked differences compared to the gaps in intermediate or higher education. For both women and men with parents from Morocco and Türkiye, differences relative to children of native-born are small at the lower end of the parental income distribution but increase towards the higher end. However, children of high-income native-born parents are more likely to complete a higher education compared to children of high-income parents from Morroco or Türkiye.<sup>30</sup>

Differences for women and men with parents from the Antilles and Surinam are smaller relative to children of native-born, and as we saw in the income plots, the gap turns positive for women and men with Antillean parents in the highest income ranks. Women and men with parents from Indonesia are somewhat more likely to have completed a higher education than children of native-born, for all income groups. Also for all income groups, both women and men with parents from

<sup>&</sup>lt;sup>28</sup>This is consistent with the findings of Chetty et al. [2020] of small gaps in college attendance rates for Black and white daughters in the US (and somewhat larger gaps for Black and white sons).

<sup>&</sup>lt;sup>29</sup>See also Figure A.8 in the Supplementary Material.

<sup>&</sup>lt;sup>30</sup>This is likely related to the multitrack educational system for secondary school in the Netherlands [Kalmijn and Kraaykamp, 2003] and inequality in primary school teachers' track recommendations [Statistics Netherlands, 2022].

Table 4: Predicted Average Outcomes Children at the 25th and 75th Percentile Income Rank of Parents

Nati	ive-born	Turl	kish	Moro	ccan	Surine	umese	Anti	llean	Indor	esian	Chi	nese	
	Level	Level	%-diff	Level	%-diff	Level	%-diff	Level	%-diff	Level	%-diff	Level	%-diff	
Panel A: Men														
25th Percentile														
Primary Income Rank	57.1	51.7	-9.5	44.7	-21.7	46.2	-19.1	43.6	-23.7	52.3	-8.5	60.2	5.3	
Share Intermediate or Higher Educated	86.5	80.1	-7.4	79.4	-8.2	82.6	-4.4	78.2	-9.6	86.4	-0.1	89.8	3.9	
Employment Rate	94.2	89.4	-5.1	80.6	-14.5	87.8	-6.9	85.4	-9.4	92.5	-1.9	96.0	1.9	
Hours per Week	34.7	30.5	-12.1	25.4	-26.8	29.0	-16.3	27.7	-20.0	32.3	-6.9	34.6	-0.1	
Hourly Wage Rank	46.9	46.6	-0.5	47.2	0.8	43.2	-7.9	42.5	-9.3	45.1	-3.8	53.7	14.6	
75th Percentile														
Primary Income Rank	67.3	61.4	-8.8	57.8	-14.1	60.1	-10.7	61.5	-8.7	63.5	-5.6	67.7	0.6	
Share Higher Educated	47.1	32.0	-32.0	36.0	-23.5	43.0	-8.8	47.3	0.3	53.1	12.6	69.5	47.6	
Employment Rate	97.7	93.8	-4.0	91.9	-6.0	94.9	-2.9	95.6	-2.1	96.6	-1.2	98.1	0.5	
Hours per Week	36.7	34.2	-6.9	32.0	-13.0	33.7	-8.2	34.2	-6.8	35.1	-4.5	36.0	-2.0	
Hourly Wage Rank	58.0	55.2	-4.8	57.2	-1.4	54.6	-6.0	56.4	-2.8	56.1	-3.3	63.6	9.7	
rattel D: Women 9545 Domontilo														
			2 1 7		ç T	0.00	c c		ې ۲		1 1	5 0 0	1 0 1	
Frimary Income Rank	34.7	28.6	-17.6	30.7	-11.6	33.9	-2.3	30.7	-11.6	37.4	7.7	53.0	52.7	
Share Intermediate or Higher Educated	88.6	86.9	-1.9	88.0	-0.6	88.5	-0.1	86.7	-2.1	89.9	1.5	93.7	5.8	
Employment Rate	88.8	75.5	-14.9	68.0	-23.4	81.1	-8.6	79.4	-10.5	88.3	-0.5	93.1	4.9	
Hours per Week	24.0	18.6	-22.3	17.3	-27.6	21.4	-10.5	18.9	-21.1	25.2	5.2	30.6	27.8	
Hourly Wage Rank	42.8	39.2	-8.4	45.0	5.3	40.8	-4.7	39.0	-8.8	40.5	-5.2	53.3	24.7	
75th Percentile														
Primary Income Rank	46.5	37.9	-18.5	41.1	-11.7	46.2	-0.6	47.7	2.6	49.4	6.2	54.6	17.5	
Share Higher Educated	57.0	45.2	-20.7	47.1	-17.5	54.5	-4.4	59.3	4.1	62.5	9.6	76.6	34.4	
Employment Rate	95.4	85.4	-10.5	79.4	-16.8	91.8	-3.8	92.6	-3.0	94.4	-1.0	94.0	-1.5	
Hours per Week	28.0	23.2	-17.0	22.2	-20.6	27.2	-2.8	24.6	-11.9	29.0	3.6	31.2	11.6	
Hourly Wage Rank	53.7	47.8	-11.1	52.3	-2.6	51.0	-5.0	52.7	-1.9	52.9	-1.6	59.2	10.2	
NOTES: This table gives the predicted personal of	sons and d	aughters	at the 25th	and 75tl	n percentil	e of the d	isposable	househole	d income r	ank of th	e parents,	using a li	near	
regression. The personal outcomes reported are th	ne mean pri	mary incc	me (wage	s and prof	its, includ	ing the ze	ros) rank,	the shar	e that has	complete	d tertiary	education	ı, the	
employment rate (wage and profit income unequa	l to zero), n	rean hour	s worked <sub>1</sub>	er week l	y employ	es (for th	ie self-em	oloyed we	do not ob	serve the	hours) and	d the mea	an hourly we	ıge rank (of

employees) of the children at the 25th and 75th percentile rank of the parents.

China are (far) more likely to have completed a higher education than natives, for all income groups and in particular for women from parents with a relatively low income.

Overall, we see a correspondence between differences in outcomes in personal primary incomes and differences in the highest completed level of education. However, what is also interesting to note is that for all groups, differences in educational outcomes between women and men are much less pronounced than the differences in primary incomes. Actually, for all groups and across parental income ranks, women are typically more likely to have completed a higher education than men, whereas their primary incomes are typically lower than that of men from the same group. Hence, for all groups, including natives, gender gaps emerge between finishing initial education and the labour market.<sup>31</sup> Also note, that this gender gap in primary incomes appears to be the smallest for children of Chinese parents.<sup>32</sup>

**Employment Rates** We now turn to the differences in labour market outcomes. Regarding employment rates, men with parents from Morocco, Türkiye, the Antilles, and Surinam are much less likely to be employed than sons of native-born, though the gap is smaller for children of high-income parents.<sup>33</sup> Differences for men with parents from Indonesia and China are small, with employment rates for men with parents from China being slightly higher at the lower end of the parental income distribution, and slightly lower for men with Indonesian parents across the entire distribution. Employment rates for women with parents from Morocco and Türkiye are much lower than for daughters of native-born, in particular for those with parents from Morocco. The gender gap in employment rates between men and women from Morocco and Türkiye is also the largest, at least for the countries of origin we consider here. Social and cultural norms regarding the division of work and care within households may play a role here. Employment rates for women with parents from the Antilles and Surinam are also substantially lower than for daughters of native-born at the lower end of the parental income distribution, though the gap disappears at the higher end. Employment rates for women with parents from Indonesia are again close to natives (though the gap is negative), whereas employment rates for women with parents from China are higher than for daughters of native-born with lower-income parents, but the gap turns negative for daughters of

 $<sup>^{31}</sup>$ Note that these differences may be particularly pronounced at the ages at which we measure the incomes and outcomes for children, as these are also the ages when many of them have children.

 $<sup>^{32}</sup>$ Chetty et al. [2020, Online Appendix Table VIII] also find relatively high college attendance rates with a small gender gap for Asian children.

<sup>&</sup>lt;sup>33</sup>See also Figure A.9 in the Supplementary Material.

higher-income parents.

Hours Worked per Week The differences in hours worked per week generally follow the same qualitative patterns as the differences in employment rates.<sup>34,35</sup> However, differences in hours worked per week are typically larger than differences in employment rates in percentage terms, see Table 4. In particular, women with parents from Morocco and Türkiye work fewer hours per week than daughters of native-born. On the other hand, women with parents from China work more hours per week than daughters of native-born (in particular for low-income parents), which partly explains their relatively high income levels.

**Hourly Wages** The results are more mixed for differences in hourly wage ranks.<sup>36</sup> For men with parents from Morocco and Türkiye, there are almost no gaps relative to sons of native-born at the 25th percentile of parental income in terms of the hourly wage rank, with a slight negative gap at the 75th percentile, but much smaller than the gap for hours worked per week. Sons with parents from Surinam and the Antilles have the largest gap in hourly wage rank of all the groups at the 25th percentile of parental income, but gaps are small for Antillean men at the 75th percentile. Differences for men with a parent from Indonesia are again small relative to sons of native-born. Men with parents from China have a positive gap in hourly wage ranks, which is likely related to having higher levels of education than sons of native-born.

For women with parents from Morocco, Türkiye, the Antilles, and Surinam the gaps in hourly wage ranks relative to daughters of native-born are typically smaller than for employment rates and hours worked per week (except for women from Türkiye and Surinam with high-income parents). The mean wage rank of women with parents from Indonesia is slightly lower than for daughters of native-born at the lower end of the distribution, but closes towards the higher end. Hourly wage ranks for women with parents from China are higher than for daughters of nativeborn, which is likely closely related to the higher educational outcomes for this group.

Overall, the most important driver of the gaps in primary income, seems to be difference in hours worked per week, though differences in employment rates and

<sup>&</sup>lt;sup>34</sup>See Figure A.10 in the Supplementary Material.

<sup>&</sup>lt;sup>35</sup>Hours worked per week and hourly wages are only observed for employees. Differences in the share of self-employed are given in Figure A.11 in the Supplementary Material. Men with parents from Morocco, Türkiye and China are notably more likely to be self-employed than natives. For women, this is only the case for those with parents from China.

 $<sup>^{36}\</sup>mathrm{See}$  also Figure A.12 in the Supplementary Material.

hourly wage ranks also play a role.<sup>37</sup> For the groups that close the gaps in primary income, such as men with parents from China, the most important driver seems to be hourly wages, which is closely related to higher educational outcomes.<sup>38</sup> Among women, those with parents from Indonesia have a positive primary income gap that is mostly driven by hours worked per week, not by hourly wage rank, despite their higher education. For women with parents from China, the positive gap in primary incomes is driven by both higher wage ranks and more hours worked per week, which may be related to their higher educational outcomes relative to women with native-born parents.<sup>39</sup> The mean wage rank of women with parents from Indonesia is lower than for daughters of native-born, despite being more educated at both the 25th and the 75th percentile.

While educational outcomes are important for closing the income gaps, Table 4 suggests there might be inequality in the returns to education for these immigrant groups, especially for men. At the 75th percentile, men with parents born in the Antilles have lower employment rates, lower hourly wages, and lower weekly hours worked, leading to lower primary incomes, despite being equally educated as sons of native-born. For Indonesian men, the gaps are smaller, but we see a similar pattern despite being slightly higher educated than the sons of native-born. Chinese men are almost 50% more likely then sons of native-born to have completed a higher education, but this barely translates to higher primary incomes. This suggests that there are barriers (i.e. discrimination) in the Dutch labor market that lead to inequality in the returns to education for second generation migrants from the countries studied. For women, this pattern is less pronounced, though daughters of Antillean parents experience negative hourly wage gaps, despite higher educational levels at the 75th percentile of the parental income distribution as well.<sup>40</sup>

#### 6 Conclusions

In this paper we have studied intergenerational mobility for natives and immigrants, with a focus on immigrants from lower-income countries, using administrative data on the universe of individuals and households in the Netherlands. We find that the

<sup>&</sup>lt;sup>37</sup>Chetty et al. [2020, Figure VI] find small differences in these outcomes for Black and white daughters, whereas differences for men seem primarily driven by differences in employment rates, though differences in hours worked and hourly wages (ranks) also play a role.

<sup>&</sup>lt;sup>38</sup>Chetty et al. [2020, Online Appendix Table VIII] also find that differences in hourly wages are the main driver of differences in primary incomes for Asian sons.

<sup>&</sup>lt;sup>39</sup>Chetty et al. [2020, Online Appendix Table VIII] find that differences in hourly wages are the main driver of differences in primary incomes for Asian daughters.

<sup>&</sup>lt;sup>40</sup>See also Figure A.7, A.8, A.10 and A.12 in the Supplementary Material.

absolute income mobility is typically much lower for immigrants, and the relative income mobility is typically somewhat lower for immigrants as well. Based on our results, we predict large and persistent income gaps for future generations. However, we also uncover substantial differences by country of origin and gender. Indeed, children with parents from Morocco and Türkiye have notably lower household incomes than natives, while children with parents from China have higher household incomes than natives. Part of this is due to differences in household composition (with immigrant children being more likely to be single, and when they are in a couple, more likely to be with a partner who also has parents from a lower-income country and also tends to face negative income gaps) and part of this is due to differences in personal incomes. We show that the latter is closely related to the differences in the level of education of the children. In terms of labor market outcomes, we tend to find larger differences in hours worked than in employment rates, where the gap in these outcomes goes in the same direction as the gap in primary personal incomes, and more mixed results for differences in hourly wages. For all countries of origin, women typically have higher educational outcomes than men, but the labour outcomes tend to be worse for women than for men for each country of origin of the parents, in particular in terms of employment rates and hours worked per week.

Our results suggest that income differences between future generations of immigrants and natives are likely to persist. We have also explored which factors may play a role in these persistent differences. Interesting directions for future research remain on what can be done to improve the outcomes for (future generations of) immigrants and underlying factors. Differences in educational outcomes seem to play a key role. Even so, education only 'explains' part of the labor market differences we find. Indeed, gaps in outcomes on the labour market are also driven by differences in the opportunities children get, by differences in norms and preferences, and by potential barriers (i.e. discrimination) which likely play an important role in the heterogeneity in labour market outcomes we find, by country of origin and gender. This too remains an important topic for future research. Finally, it would also be interesting to study the role of the neighbourhood in which children of immigrants and natives grow up [following e.g. Chetty et al., 2020, Bratu and Bolotnyy, 2023], although the contrast in opportunities across neigborhoods is likely to be less pronounced in the Netherlands than in e.g. the US.

#### References

- R. Abramitzky, L. Boustan, E. Jácome, and S. Pérez. Intergenerational mobility of immigrants in the United States over two centuries. <u>American Economic Review</u>, 111(2):580–608, 2021.
- P. Acciari, P. Aberto, and G. Violante. "and yet it moves": intergenerational mobility in Italy. <u>American Economic Journal, Applied Economics</u>, 14(3):118– 163, 2022.
- A. Adermon, M. Lindahl, and M. Palme. Dynastic human capital, inequality, and intergenerational mobility. American Economic Review, 111(5):1523–1548, 2021.
- A. Aydemir, W.-H. Chen, and M. Corak. Intergenerational earnings mobility among children of Canadian immigrants. <u>Review of Economics and Statistics</u>, 91(2):377– 397, 2009.
- G. Becker and N. Tomes. An equilibrium theory of the distribution of income and intergenerational mobility. Journal of Political Economy, 8(6):1153–1189, 1979.
- J. Blanden. Intergenerational income persistence. IZA World of Labor, 176, 2019.
- G. Borjas. The intergenerational mobility of immigrants. <u>Journal of Labor</u> Economics, 11(1):113–135, 1993.
- C. Bratu and V. Bolotnyy. Immigrant intergenerational mobility: A focus on childhood. European Economic Review, 151:104353, 2023.
- R. Chetty, N. Hendren, P. Kline, and E. Saez. The effects of neighborhoods on intergenerational mobility i: Childhood exposure effects. <u>Quarterly Journal of</u> <u>Economics</u>, 129(4):1553–1623, 2014.
- R. Chetty, N. Hendren, M. R. Jones, and S. R. Porter. Race and economic opportunity in the united states: An intergenerational perspective. <u>Quarterly Journal</u> of Economics, 135(2):711–783, 2020.
- M. Corak. The Canadian geography of intergenerational income mobility. Economic Journal, 130(631):2134–2174, 2020.
- N. Deutscher and B. Mazumder. Intergenerational mobility across Australia and the stability of regional estimates. Labour Economics, 66:101861, 2020.
- M. Hammarstedt and M. Palme. Human capital transmission and the earnings of second-generation immigrants in Sweden. IZA Journal of Migration, 1(4), 2012.

- S. Heidrich. Intergenerational mobility in sweden: a regional perspective. <u>Journal</u> of Population Economics, 30(4):1241–1280, 2017.
- M. Kalmijn and G. Kraaykamp. Dropout and downward mobility in the educational career: An event-history analysis of ethnic schooling differences in the netherlands. Educational Research and Evaluation, 9(3):104974, 2003.
- G. Kenedi and L. Sirugue. Intergenerational income mobility in France: A comparative and geographic analysis. Journal of Public Economics, 226:104974, 2023.
- R. Manduca, M. Hell, A. Adermon, J. Blanden, E. Bratberg, A. Gielen, H. Van Kippersluis, K. Lee, S. Machin, M. Munk, M. Nybom, Y. Ostrovsky, S. Rahman, and O. Sirniö. Measuring absolute income mobility: lessons from North America and Europe. American Economic Journal: Applied Economics, 16(2):1–30, 2024.
- H. Nicolaas and A. Sprangers. Buitenlandse migratie in nederland 1795–2006: de invloed op de bevolkingssamenstelling (in Dutch). <u>Bevolkingstrends</u>, 2007(4): 32–46, 2007.
- OECD. Catching Up? Intergenerational Mobility and Children of Immigrants. OECD Publishing, OECD, Paris, 2017.
- OECD. Under Pressure: The Squeezed Middle Class. OECD publishing, OECD, Paris, 2019.
- G. Solon. Intergenerational mobility in the labor market. In O. Ashenfelter and D. Card, editors, <u>Handbook of Labor Economics</u>, volume 3, chapter 10, pages 1761–1800. Elsevier, Amsterdam, 1999.
- Statistics Netherlands. Meten van inkomen en inkomensongelijkheid (in Dutch), 2018a. Centraal Bureau voor de Statistiek.
- Statistics Netherlands. Integratie en Samenleven 2022 (in Dutch), 2022. Statistics Netherlands.
- L. Stoeldraaijer, C. Van Duin, H. Nicolaas, and C. Huisman. Kernprognose 2022-2070: Door oorlog meer migranten naar Nederland (in Dutch). <u>Statistische</u> Trends, 2022.

# Intergenerational Mobility of Immigrants in the Netherlands

# Supplementary Material

Roel van Elk Egbert Jongen

Patrick Koot Alice Zulkarnain

May 2024

#### A.1 Supplementary tables

	Native-born	Immigrant	Higher-income Country	Lower-income Country	Moroccan	Turkish	Surinamese	Antillean	Indonesian	Chinese
1. Median household income parents	48,925	36,509	43,589	33,188	32,223	32,453	34,359	33,915	47,268	34,236
2. Median rank parents (household income)	54.0	27.0	42.0	21.0	19.0	20.0	23.0	22.0	51.0	23.0
3. Median household income child	46,395	38,592	42,731	36,080	32,483	39,451	34,398	33,374	43,881	44,720
4. Median rank child (household income)	53.0	37.0	45.0	32.0	26.0	38.0	29.0	28.0	47.0	49.0
5. Median equivalized househ. inc. parents	25,073	19,466	23,361	17,610	15,665	16,902	19,827	19,347	25,124	17,759
6. Median rank parents (stand. househ. inc.)	54.0	26.0	46.0	18.0	12.0	16.0	28.0	26.0	54.0	19.0
7. Median equivalized household inc. child	31,014	26,867	29,692	25,112	22,148	25,051	25,167	23,987	30,598	33,122
8. Median rank child (stand. household inc.)	53.0	36.0	47.0	30.0	20.0	30.0	30.0	26.0	51.0	61.0
9. Median personal income child	34,637	29,901	33,013	27,955	23,829	27,082	28, 320	26,767	34,137	38,831
10. Median rank child (personal income)	52.0	41.0	48.0	36.0	27.0	34.0	37.0	34.0	51.0	62.0
11. Number of observations	536, 285	100,178	36,627	63,551	11,937	13,062	14,847	5,717	7,137	1,584
NOTES: This table gives summary statistics of	î key variables	in the analysis	s. Mean househole	d income is equiv	alized dispose	uble househ	old income. M	ean persona	l income is	
personal primary income from wages and profi	its (including t	he zeros). Eur	co values represen	t 2019 euros. The	e income ranh	ts are the r	anks in the ful	l estimation	sample of pare	ents
and children, respectively. Single (parent) is d	letermined by 1	the set of indiv	viduals living in t	ne same househol	d. Higher edu	ıcated is d€	fined as ISCE	D 6 or highe	r. Employed is	
defined as having non-zero wages and profits.	Hours worked	and hourly we	ages are for emplo	yees only (so exc	luding the no	n-employed	l and the self-e	mployed). 7	The hourly wag	es
ranks are the ranks in the full estimation sam	ple of children.	Outcomes for	r parents are aver	ages for 2003–200	15, outcomes	for children	are averages f	for 2017–201	9. Observation	S

Variables
- Income
Sample -
Estimation
Children
<sup>2</sup> arents and
Statistics I
Descriptive
Table A.1: ]

are the number of child-parental household pairs.

n Immigrant	Higher-income Country	Lower-income Country	Moroccan	Turkish	Surinamese	Antillean	Indonesian	Chinese
.8 26.4	29.3	24.7	20.1	23.8	25.7	25.0	30.4	32.5
32 18.12	18.57	17.85	17.77	17.35	17.63	17.65	18.65	19.74
.0 44.0	48.0	42.0	43.0	39.0	41.0	41.0	48.0	56.5
0.86	0.91	0.83	0.72	0.82	0.86	0.85	0.93	0.95
0.88 0.88	0.90	0.86	0.83	0.83	0.87	0.85	0.92	0.92
4 0.38	0.45	0.34	0.28	0.27	0.33	0.31	0.50	0.64
.2 4.1	3.9	4.2	5.2	4.3	3.7	3.8	3.8	4.1
.3 2.3	2.2	2.4	2.6	2.7	2.3	2.3	2.2	2.1
0 0.24	0.19	0.27	0.20	0.17	0.16	0.38	0.03	0.22
4 0.31	0.25	0.34	0.20	0.24	0.50	0.49	0.24	0.20
8 100,342	36,627	63,588	11,960	13,058	14,864	5,720	7,140	1,589
es in the analysis	s. Mean household	l income is equiva	alized disposa	ble househ	old income. M	ean personal	income is	
g the zeros). Eu	ro values represent	2019 euros. The	e income rank	cs are the ra	anks in the ful	l estimation	sample of pare	nts
y the set of indiv	viduals living in th	ie same househol	d. Higher edı	ıcated is de	fined as ISCE	D 6 or highe	r. Employed is	
d and hourly wa	ges are for employ	ees only (so excl <sup>1</sup>	uding the nor	1-employed	and the self-er	nployed).		
n sample of child vairs.	dren. Outcomes fo	r parents are ave	srages for 200	3–2005, out	comes for child	dren are ave	rages for 2017–	2019.
an d x 3 3 3 2 4 4 7 5 3 8 8 8 4 4 7 5 5 7 8 4 1 4 1 4 7 6 5 7 8 4 1 4 1 4 7 6 7 7 7 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	Immigrant 26.4 $26.4$ 18.12 $44.0$ $0.86$ $0.86$ $0.88$ $0.38$ $0.38$ $0.38$ $0.33$ $0.24$ $100,342$ in the analysi the zeros). Eu the set of individual wa and hourly wa sample of chilirs.	ImmigrantHigher-incomeCountryCountry $26.4$ $29.3$ $18.12$ $18.57$ $44.0$ $48.0$ $0.86$ $0.91$ $0.86$ $0.91$ $0.88$ $0.91$ $0.38$ $0.91$ $0.38$ $0.91$ $0.38$ $0.91$ $0.31$ $0.24$ $0.19$ $0.24$ $0.24$ $0.19$ $0.21$ $0.23$ $100, 342$ $36.627$ in the analysis. Mean householdthe set of individuals living in thand hourly wages are for employsample of children. Outcomes forins.	ImmigrantHigher-incomeLower-incomeCountryCountry24.7 $26.4$ $29.3$ $24.7$ $18.12$ $18.57$ $17.85$ $44.0$ $48.0$ $42.0$ $0.86$ $0.91$ $0.83$ $0.86$ $0.91$ $0.83$ $0.86$ $0.91$ $0.83$ $0.86$ $0.91$ $0.83$ $0.38$ $0.91$ $0.83$ $0.38$ $0.91$ $0.83$ $0.38$ $0.91$ $0.32$ $0.38$ $0.91$ $0.34$ $0.31$ $0.19$ $0.24$ $0.24$ $0.19$ $0.27$ $0.31$ $0.22$ $0.34$ $100,342$ $36.627$ $63.588$ in the analysis. Mean household income is equivither eart of individuals living in the same householand hourly wages are for employees only (so excl sample of children. Outcomes for parents are averities.	Immigrant Higher-income Lower-income Moroccan   Country Country Country 20.1   26.4 29.3 24.7 20.1   18.12 18.57 17.65 17.77   44.0 48.0 0.91 0.83 0.72   0.86 0.91 0.83 0.72   0.88 0.90 0.86 0.83   0.38 0.91 0.83 0.72   0.38 0.91 0.86 0.83   0.38 0.91 0.86 0.83   0.38 0.91 0.86 0.83   0.39 0.45 0.34 0.28   0.31 0.32 0.34 0.20   0.19 0.19 0.27 0.24 2.6   100,342 36,627 63,588 11,960   in the analysis 0.23 0.34 0.20   in the set of individuals living in the same household. Higher edu and hourly wages are for employees only (so excluding the nor   insthe set of individuals	Immigrant Higher-income Moroccan Turkish   Country Country Country 20.1 23.8   26.4 29.3 24.7 20.1 23.8   18.12 18.57 17.85 17.77 17.35   18.12 18.57 17.85 17.77 17.35   0.86 0.91 0.83 0.82 0.83   0.88 0.90 0.86 0.83 0.83   0.88 0.91 0.86 0.83 0.83   0.88 0.99 0.86 0.83 0.83   0.88 0.945 0.34 0.28 0.27   0.39 0.34 0.28 0.21 0.21   100.342 0.19 0.27 0.20 0.17   0.24 0.29 10.30 1.4 2.6 2.7   0.31 0.32 0.32 0.24 2.6 2.7   0.32 0.34 0.23 0.20 0.17 0.24   100	$\begin{tabular}{ l l l l l l l l l l l l l l l l l l l$	$\begin{tabular}{ l   l  l  l  l  l  l  l  l  l  l  l  l$	$\begin{tabular}{ l l l l l l l l l l l l l l l l l l l$

#### A.2 Supplementary figures



Figure A.1: Distribution of Parent Household Income Ranks by Immigrant Group

*Notes:* Frequency distribution of equivalized disposable household income ranks of parents by immigrant group.

Figure A.2: Relationship Parent Household Income Ranks–Parent Household Wealth Ranks: Selected Countries

![](_page_45_Figure_1.jpeg)

*Notes:* Estimated linear relationship between the equivalized disposable household income rank of the parents and their own total wealth rank. Wealth data is only available starting 2006, therefore the relationship is estimated between average income ranks for 2003–2005 and average total wealth for 2006–2008.

![](_page_46_Figure_0.jpeg)

Figure A.3: Relationship Parent Household Income Ranks–Parent Total Household Wealth: Selected Countries

*Notes:* Estimated linear relationship between the equivalized disposable household income rank of the parents and their own total wealth. Wealth data is only available starting 2006, therefore the relationship is estimated between average income ranks for 2003–2005 and average total wealth for 2006–2008.

![](_page_47_Figure_0.jpeg)

Figure A.4: Relationship Non-Equivalized Disposable Household Income Parent — Child

*Notes:* Estimated linear relationship between the non-equivalized disposable household income rank of the children and the non-equivalized disposable household income rank of the parents.

![](_page_48_Figure_0.jpeg)

Figure A.5: Relationship Household Income Parents – Primary Incomes Children Born in the Netherlands

*Notes:* Estimated linear relationship between the equivalized disposable household income rank of the parents and the personal primary income rank of the children.

![](_page_49_Figure_0.jpeg)

Figure A.6: Relationship Household Income Parent – Personal Gross Income Children

*Notes:* Estimated linear relationship between the equivalized disposable household income rank of the parents and the personal gross income rank of the children.

![](_page_50_Figure_0.jpeg)

Figure A.7: Relationship Income Parent – Children Intermediate or Higher Educated

*Notes:* Estimated linear relationship between the equivalized disposable household income rank of the parents and the probability that the child has a secondary or higher level of education (ISCED 3 or higher).

![](_page_51_Figure_0.jpeg)

Figure A.8: Relationship Household Income Parent – Children Higher Educated

*Notes:* Estimated linear relationship between the equivalized disposable household income rank of the parents and the probability that the child has a tertiary education (ISCED 6 or higher).

![](_page_52_Figure_0.jpeg)

Figure A.9: Relationship Household Income Parent – Employment Rate Children

*Notes:* Estimated linear relationship between the equivalized disposable household income rank of the parents and employment rates of the children (defined as having non-zero wage and/or profit income).

![](_page_53_Figure_0.jpeg)

Figure A.10: Relationship Household Income Parent – Weekly Hours Worked Children

*Notes:* Estimated linear relationship between the equivalized disposable household income rank of the parents and hours worked per week by children (employees only).

![](_page_54_Figure_0.jpeg)

Figure A.11: Relationship Household Income Parent – Child Self-Employed

*Notes:* Estimated linear relationship between the equivalized disposable household income rank of the parents and the probability that a child is self-employed.

![](_page_55_Figure_0.jpeg)

Figure A.12: Relationship Household Income Parent – Hourly Wage Rank Children

*Notes:* Estimated linear relationship between the equivalized disposable household income rank of the parents and the hourly wage rank of children (employees only).